Author's response to reviews

Title: The adoption of mediterranean diet attenuates the development of acute coronary syndromes in people with the metabolic syndrome

Authors:

Dr Christos Pitsavos (d.b.panagiotakos@usa.net)
Dr Demosthenes Panagiotakos (d.b.panagiotakos@usa.net)
Christina Chrysohoou (chrysohoou@usa.net)
Ioanna Papaioannou (chrysohoou@usa.net)
Lambros Papadimitriou (chrysohoou@usa.net)
Dimitrios Tousoulis (chrysohoou@usa.net)
Pavlos Toutouzas (chrysohoou@usa.net)
Christodoulos Stefanadis (chrysohoou@usa.net)

Version: 5 Date: 17 Mar 2003

Dear Editor

We made all the requested changes, according to the Instructions of the Journal

Thank you once again for your interest on our work...

Sincerely

D B Panagiotakos